

Author's response to reviews

Title: Effects Of Testicular Microlithiasis On Doppler Parameters: Report Of Three Cases

Authors:

Ramazan Kutlu (rkutlu@inonu.edu.tr)

Ahmet Sigirci (asigirci@inonu.edu.tr)

Tamer Baysal (tbaysal@inonu.edu.tr)

Alpay Alkan (aalkan@inonu.edu.tr)

Kaya Sarac (ksarac@inonu.edu.tr)

Version: 4 **Date:** 26 Dec 2001

Dear Editor,

We have revised our manuscript. We changed the title and the text according to the reviewer's comments. Figures with better resolution submitted.

We'd like to thank you for considering our manuscript.

Sincerely yours,

Ramazan Kutlu, MD